
Daily Status - 9/14/16
9/14/16

Summary: Light winds,  < 5 m/s, out of the WSW, clear,
                 Worked on new DSM to test questionable GILL 2D cable. To verify
                 DSM was setup correctly built a test cable for the GILL. Test cable
                 verified DSM was setup correctly so 2D cable is bad. Will take back to Boulder            

Actions past 24 hours:

 Base work, no boating yesterday
build test cable to check GILL 2D and DSM settings

To dos:

 move Mote #1 and NR01 to site 9

add soil probes to Mote #1

Sensor Status: 

T/RH:  ok
P: ok
2D Gill: ok
csat u,v:  ok
csat ldiag: ok
csat w, tc: ok
EC150: ok
motes: ok, mote #1 does not have soil probes
Wetness: ok
radiation: ok
Tsoil: ok
Gsoil: ok
Qsoil: ok
Cvsoil: not responding
Vbatt: ok
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